Author’s response to reviews

Title: Effects and outcomes of septostomy in twin-to-twin transfusion syndrome after fetoscopic laser therapy

Authors:

Wen-Fang Li (wanna0507@gmail.com)
An-Shine Chao (aschao1295@cgmh.org.tw)
Shuenn-Dyh Chang (gene@cgmh.org.tw)
Po-Jen Cheng (pjcheng@cgmh.org.tw)
Lan-Yan Yang (lyyang0111@gmail.com)
YL Chang (j12054@cgmh.org.tw)

Version: 3 Date: 08 Oct 2019

Author’s response to reviews:

1. We note that the current submission contains some textual overlap with other previously published works, in particular:


Overlap occurs throughout the methods section.

Since the overlap is contained within the Methods section, and rephrasing the methods would not be conducive to future reproducibility of your study, we ask that, for these sections of textual overlap, you provide a summarizing statement explaining this methodology has been published previously and provide the attribution to the sources mentioned above.

Answer: thanks for comment and we have changed these sentences as “The procedure of laser therapy for TTTS in our hospital has been described in a previous study [12].”

2. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Answer: Thanks, we had done it accordingly.